Author's response to reviews

Title: Evaluation of CXCL9 and CXCL10 as circulating biomarkers of human cardiac allograft rejection.

Authors:

Kristjan Karason (kristjan.karason@medfak.gu.se)
Margareta Jernas (Margareta.jernas@medic.gu.se)
Daniel A Hagg (Daniel.hagg@medic.gu.se)
Per-Arne Svensson (Per-Arne.Svensson@medic.gu.se)

Version: 3 Date: 29 May 2006

Author’s response to reviews: see over
Reviewer: Marja Steenman
Reviewer's report:
General

-------------------------------------------------------------------------------

Major Compulsory Revisions (that the author must respond to before a decision on publication can be reached)

-------------------------------------------------------------------------------

Minor Essential Revisions (such as missing labels on figures, or the wrong use of a term, which the author can be trusted to correct)

The authors state in their response that for cluster 3 no genes were omitted based on absent/present call

Answer: The following sentence has been added to the text
“All of the genes in cluster 3 were classified as detectable by the detection call algorithm.” (page10, section 3, line 8)

The expression profiles presented in this manuscript has been submitted to Gene Expression Omnibus. The code for this dataset is GSE4315. This has is indicated in the manuscript (page 10, section 2, line 4)